---
title: Orø -- vinenes ø
date: 2024-12-11
author:
  - Birgit Friis Petersen
image: ../vintaarn/vinothek.jpg
categories: [handel]
filters:
  - footer
---

{{< include _index.qmd >}}

[Marlieses og Claus' vine](../birkevin/index.qmd)  
[Bings og Provianten](../biovin/index.qmd)  
[Vinskuret](../vinskuret/index.qmd)  
[Vintårnet på Langdalgaard](../vintaarn/index.qmd)  
[Kristians honningvin](../honningvin/index.qmd)
md?h=issue4' type='application/atom+xml'/>
<link rel='vcs-git' href='https://source.couchdesign.dk/posten' title='posten Git repository'/>
<link rel='vcs-git' href='user@source.couchdesign.dk:posten' title='posten Git repository'/>
</head>
<body>
<div id='cgit'><table id='header'>
<tr>
<td class='logo' rowspan='2'><a href='/'><img src='/cgit-css/cgit.png' alt='cgit logo'/></a></td>
<td class='main'><a href='/'>index</a> : <a href='/posten/'>posten</a></td><td class='form'><form method='get'>
<input type='hidden' name='id' value='8b3d143e7e917a8be4ec1a324a4a71642b4a57de'/><select name='h' onchange='this.form.submit();'>
<option value='main'>main</option>
</select> <input type='submit' value='switch'/></form></td></tr>
<tr><td class='sub'>Unnamed repository; edit this file 'description' to name the repository.</td><td class='sub right'>Jonas Smedegaard</td></tr></table>
<table class='tabs'><tr><td>
<a href='/posten/?h=issue4'>summary</a><a href='/posten/refs/?h=issue4&amp;id=8b3d143e7e917a8be4ec1a324a4a71642b4a57de'>refs</a><a href='/posten/log/nummer4/halloween/_index.qmd?h=issue4'>log</a><a class='active' href='/posten/tree/nummer4/halloween/_index.qmd?h=issue4&amp;id=8b3d143e7e917a8be4ec1a324a4a71642b4a57de'>tree</a><a href='/posten/commit/nummer4/halloween/_index.qmd?h=issue4&amp;id=8b3d143e7e917a8be4ec1a324a4a71642b4a57de'>commit</a><a href='/posten/diff/nummer4/halloween/_index.qmd?h=issue4&amp;id=8b3d143e7e917a8be4ec1a324a4a71642b4a57de'>diff</a></td><td class='form'><form class='right' method='get' action='/posten/log/nummer4/halloween/_index.qmd'>
<input type='hidden' name='h' value='issue4'/><input type='hidden' name='id' value='8b3d143e7e917a8be4ec1a324a4a71642b4a57de'/><select name='qt'>
<option value='grep'>log msg</option>
<option value='author'>author</option>
<option value='committer'>committer</option>
<option value='range'>range</option>
</select>
<input class='txt' type='search' size='10' name='q' value=''/>
<input type='submit' value='search'/>
</form>
</td></tr></table>
<div class='path'>path: <a href='/posten/tree/?h=issue4&amp;id=8b3d143e7e917a8be4ec1a324a4a71642b4a57de'>root</a>/<a href='/posten/tree/nummer4?h=issue4&amp;id=8b3d143e7e917a8be4ec1a324a4a71642b4a57de'>nummer4</a>/<a href='/posten/tree/nummer4/halloween?h=issue4&amp;id=8b3d143e7e917a8be4ec1a324a4a71642b4a57de'>halloween</a>/<a href='/posten/tree/nummer4/halloween/_index.qmd?h=issue4&amp;id=8b3d143e7e917a8be4ec1a324a4a71642b4a57de'>_index.qmd</a></div><div class='content'>blob: 0991442284fe0d91935dcf780d84852f680b729b (<a href='/posten/plain/nummer4/halloween/_index.qmd?h=issue4&amp;id=8b3d143e7e917a8be4ec1a324a4a71642b4a57de'>plain</a>)
<table summary='blob content' class='blob'>
<tr>
<td class='lines'><pre><code>